

Journal of Geophysical Research: Biogeosciences

RESEARCH ARTICLE

10.1029/2018JG004621

Key Points:

- The microbial diversity of brines from Salar de Atacama differs according to the concentration of lithium
- The brines with low lithium concentrations were dominated by Archaea, and Bacteria dominated lithium- concentrated brines
- Our research revealed the presence of microbial life in one of the most saline environments that has been described on Earth

Supporting Information:

• Supporting Information S1

Correspondence to: C. Dorador, cristina.dorador@uantof.cl

Citation:

Cubillos, C. F., Aguilar, P., Grágeda, M., & Dorador, C. (2018). Microbial communities from the world's largest lithium reserve, Salar de Atacama, Chile: Life at high LiCl concentrations. *Journal of Geophysical Research: Biogeosciences*, *123*, 3668–3681. https://doi.org/ 10.1029/2018JG004621

Received 23 MAY 2018 Accepted 18 NOV 2018 Accepted article online 3 DEC 2018 Published online 22 DEC 2018

Microbial Communities From the World's Largest Lithium Reserve, Salar de Atacama, Chile: Life at High LiCl Concentrations

JGR

Carolina F. Cubillos^{1,2,3} , Pablo Aguilar⁴, Mario Grágeda², and Cristina Dorador^{1,3,5}

¹Laboratorio de Complejidad Microbiana y Ecología Funcional, Instituto Antofagasta, Universidad de Antofagasta, Antofagasta, Chile, ²Department of Chemical Engineering and Mineral Process and Center for Advanced Study of Lithium and Industrial Minerals (CELIMIN), Universidad de Antofagasta, Antofagasta, Chile, ³Centre for Biotechnology and Bioengineering, Universidad de Chile, Santiago, Chile, ⁴Lake and Glacier Ecology Research Group, Deparment of Ecology, University of Innsbruck, Innsbruck, Austria, ⁵Departamento de Biotecnología, Facultad de Ciencias del Mar y Recursos Biológicos, Universidad de Antofagasta, Antofagasta, Chile

Abstract Salar de Atacama is one of the largest global reservoirs of natural lithium brines (mean lithium concentration = 1,500 ppm), enabling Chile to be a leading producer of lithium products. This large salar (3,000 km²), located in the Atacama Desert at 2,300 m above sea level, is dominated by microorganisms; however, little is known about the microbes present in the brines associated with this economically important mining process. Here we study lithium as a modulator of microbial richness and diversity in brines representing natural conditions (34.7% salinity) and conditions prior to lithium production with a concentrated brine (55.6% salinity). Brines only supported a single archaeal family (Halobacteriaceae): natural brines included the archaeal genera Halovenus, Natronomonas, Haloarcula, and Halobacterium. Concentrated brines included the archaeal genera Halovenus, Halobacterium, and Halococcus. The most abundant bacterial families in natural brine were Rhodothermaceae and Staphylococcaceae; Xanthomonadaceae dominated the bacterial community in the concentrated brine. A comparison of entire microbial community (Archaea and Bacteria) revealed that only seven operational taxonomic units were shared between samples, all of which were Archaea. Further, our results showed that Bacteria were phylogenetically more diverse and rich in the concentrated brine, while archaeal diversity was maximized in the natural brine. The concentrated lithium brines of the Salar de Atacama represent one of the most saline environments described to date (dominated by LiCl). We suggest that elevated concentrations of lithium could greatly modulate microbial diversity and give insights into the adaptive biology of microorganisms required to cope with extremely high concentrations of salts that extend beyond that of NaCl, a far more commonly studied salt.

Plain Language Summary Lithium is a main component of many of the batteries that we rely on for our daily use. In the last years, nearly 40% of all lithium obtained globally was from a single fragile salt-lake ecosystem: Salar de Atacama in the Atacama Desert of northern Chile. This salar has extremely saline waters called brines (dominated by NaCl, aka table salt), which are naturally highly concentrated in lithium and concentrated in evaporation ponds. In total, concentrated brines has a salinity of 55.6% (ocean salinity = 0.3%), representing one of the most saline environments described on Earth to date (dominated by lithium chloride). Our research has shown that concentrated brines support life and are dominated by hundreds of species of microorganisms. Due to saline stress these "extremophiles" have developed very special (and previously undescribed) strategies to survive in this lithium soup. These results have implications beyond Earth: they have marked implications for our understanding for the potential for life on Mars, where liquid water is known to occur as brine. Although lithium production has clear economic importance, our results show that we should consider how we will preserve these unusual ecosystems that act as reservoirs of unique microbial life.

1. Introduction

©2018. American Geophysical Union. All Rights Reserved. Located in the central Andes of northern Chile, the Salar de Atacama has a surface area of 3,000 km² (Lowenstein & Risacher, 2009). It was formed in the early Late Cretaceous as a consequence of the inversion of the Mesozoic backarc and is located at an elevation of 2,300 m above sea level (Mpodozis et al., 2005).



Aquatic habitats in the salar are highly saline, with brines being dominated by NaCl (Lowenstein & Risacher, 2009). Briefly, brines are defined as concentrated salt solutions that naturally occur in the form of natural coastal lagoons, salt and soda lakes, or as artificial salterns or saltworks (Trüper & Galinski, 1986) and are typically associated with NaCl. Brines from Salar de Atacama are formed under particular conditions that determine not only their physical and chemical characteristics but also the composition of its microbial communities. These saline systems include a number of different constituents: these include a liquid fraction, largely made up of chlorinated brines; a detrital fraction, consisting of sands, salts, and intertwined clays in the saline body; and a salt fraction, produced by the deposition of different salts resulting in areas dominated by carbonate, sulfate, and chlorides (Vila, 2010). Of these salts, lithium is important in the Salar de Atacama, which exhibits some of the highest reported concentrations of lithium in the world (Munk et al., 2016), with a mean concentration of 1,500 ppm (1.5 g/L) of Li (Habashi, 1997). The salar has been estimated to include total reserves of lithium in excess of 1,200,000 t (Vine, 1976).

Lithium is an alkali metallic element with a low melting point (Habashi, 1997) and is the 27th most abundant element in nature (Aral & Vecchio-Sadus, 2008; Habashi, 1997). The bulk of lithium is found in natural brines, and half of all world reserves are found in three countries in South America (Argentina, Bolivia, and Chile; Del Barco & Foladori, 2014). In recent years, extraction of lithium from natural brines has markedly increased: Chile dominates world production with the Salar de Atacama being the principal source. Here lithium is extracted by pumping brines into shallow evaporation ponds, followed by evaporation under controlled conditions that largely eliminates other components/ions (Gruber et al., 2011). In 2004, Garret showed that the chemical composition of natural brines from Salar de Atacama is dominated by sodium (91 ppm), potassium (23,600 ppm), lithium (1,570 ppm), and magnesium (9,650 ppm). The mining industry in the region takes advantage of high evaporation rates (3,700 mm/year) and solar radiation (6.3×10^6 cal·m⁻²·day), combined with low precipitation rates (10-25 mm/year), to allow the lithium extraction (Garcés, 2000; Garrett, 2004). During the initial phase of the extraction process (Figure S1), lithium is concentrated from 0.12% to 6% in solar evaporation ponds (Garcés, 2000; Garrett, 2004). This concentrated brine is subsequently transported to a chemical process plant, where lithium carbonate is obtained as a final product. Although the chemical composition of salty brines from Salar de Atacama has been extensively studied (e.g., Garcés, 2000; Garrett, 2004; Vila, 2010), little attention has been paid to microbial life associated with these brines. Particularly, studies examining the effects of high concentrations of lithium in microorganisms, as well the changes in microbial community structure in these brines during this economically important mining process of Chile are lacking.

Hypersaline environments have been widely studied, often with a focus on their saline compositions, high salinity and in some case, to understand microbial community composition. Some sites have been particularly well studied, including the Great Salt Lake (dominated by NaCl; Brock, 1975), Dead Sea (MgCl₂; Oren, 1992), Salar de Uyuni (NaCl/LiCl; Haferburg et al., 2017), and Abu Gabara Lake (Na₂CO₃; Imhoff et al., 1978). Halophile, or salt-loving organisms are classified as slight, moderate or extreme, depending in the concentrations of salts required for growth (DasSarma & DasSarma, 2012). Generally, the main bacterial groups detected in saline environments belong to Alphaproteobacteria, Gammaproteobacteria, Actinobacteria, and Bacteroidetes (Oren et al., 2014; Ventosa et al., 2015), and Euryarchaeota and Nanohaloarchaeota in the Archaea (Mora-Ruiz et al., 2018; Ochsenreiter et al., 2002). In the case of lithium brines, only Halobacteriaceae, a family within Euryarchaeota (Haferburg et al., 2017), have been reported.

The biological role of lithium remains unclear, and it has been described as an element that does not appear to be essential for life (Wackett et al., 2004). It has a remarkably high capacity to replace other cations (Na⁺, K⁺, Mg⁺², and Ca⁺²) in living organisms, due to its reduced ionic radius and high polarizing strength (Sapse & Schleyer, 1995; Shahzad et al., 2017). In the case of prokaryotes, it has been suggested that some fungal groups have resistance or tolerance mechanisms against lithium (de Assunção, 2012). Further, other effects of lithium on microbial cells have been demonstrated including the stimulation of autolysis (Sugahara et al., 1983) and sporulation (Warburg et al., 1985), as well as growth inhibition (Kurita & Funabashi, 1984; Takenishi & Takada, 1984). Conversely, lithium has been used as a growth medium for the isolation of *Straphylococcus aureus, Listeria monocytogenes*, and *Bifidobacterium* sp. (Cox et al., 1990; Lapierre et al., 1992; Minor & Marth, 1976), and evidence exists of its accumulation in bacteria, actinomycetes, fungi, and yeasts (Perkins & Gadd, 1993; Tsuruta, 2005). In addition, it has been shown that *Micrococcus varians* ssp.



Figure 1. Map of Salar de Atacama and satellite image of evaporation ponds of lithium brines concentration process (blue squares; source EarthExplorer, U.S. Geological Survey, https://earthexplorer.usgs.gov/).

halophilus, a moderate halophilic bacterium, can grow in concentrations of up to 1.5 M of lithium chloride (Kamekura & Onishi, 1982), one of the highest values reported.

Despite the growing interest in the extraction and use of lithium to produce electrical appliances, there is little knowledge regarding the composition of microbial communities present in lithium brines, which can be considered as an underexplored microbial habitat. The aims of this study are to determine and to describe the diversity and composition of Archaea and Bacteria present in natural and concentrated, highly lithiumenriched brines from Salar de Atacama. We also aimed to clarify the role of lithium as a modulator of changes in microbial diversity associated with the concentration process of brines in the Salar de Atacama. Thus, our results described the microbial community composition from one environment with high concentrations of non-NaCl salts, which may be key for approaches in the search for extraterrestrial life, for example, on other planets or celestial bodies.

2. Methods

2.1. Study Area and Sampling

The industrial process of lithium extraction from Salar de Atacama brines begins with the extraction of natural brine from a depth of 50 m (>31.5 L/s) from the saline crust of the Salar (Figure 1). These brines are subsequently deposited in storage pools, where local environmental conditions such as the high evaporation rate (3,200 mm/year), minimal average rainfall (10 mm/year) and elevated concentrations of solar radiation (6.3×10^6 cal·m⁻²·day) allows the evaporation of water and crystallization/precipitation of salts over a sequential process of 12–14 months (Garcés, 2000). Here two samples of different types of brines were analyzed: a natural brine with 0.16% of lithium and a concentrated brine from the industrial process with 8.12% of lithium. Both brines are liquid (Figure S2); however, the concentrated brine was highly viscous with a yellow hue. Five liters of each brine were collected directly from the evaporation pools (initial and final stage respectively) in August 2014. Previous studies have shown that the different evaporation pools exhibit a growing salt concentration leading by lithium (Garrett, 2004). Due to the restricted access (only two main companies produce lithium in Chile) and sampling difficulties in the Salar de Atacama, it was unfortunately not possible to sample the entire process. However, this is one of the first descriptions of microbial communities associated with lithium evaporation ponds.



2.2. Characterization of Brines

Both brines were stored at 4 °C in the dark until processing. Lithium, magnesium, sodium, potassium, and calcium were measured by direct aspiration flame-air atomic absorption spectrophotometry (Model220FS-Varian, Inc.). Chloride measurements were made through the argentometric method, and finally, sulfate by sulfate determination through drying residue. Furthermore, we measured acidity (acid-base volumetric analysis), conductivity (Metrohm conductivity-five rings), density (Anton Paar, model DSA 5000 M), pH (781 pH/Ion Meter-Metrohm), salinity (absolute salinity calculated from chemical analysis), surface tension (Lauda TD-3), total dissolved salts (dried at 180 °C-SM 2540 C), total organic carbon (SM 5310 C), turbidity (Turbidimeter Zhifong, model WGZ-B), viscosity (viscometer AVS 310 Schott Gerate), and water activity (Novasina aw center, model AWC503-C; American Public Health Association, 1915). In parallel, we measured the presence of organic solvents, hydrocarbons, and oils in both brines (SM 5220 B).

2.3. DNA Extraction and 16S rRNA Gene Massive Sequencing

Two treatments were used to obtain genomic DNA from brine samples. The first consisted of filtering 50 mL of both brines through 0.22-um filters (Supor 200 Membrane Disc Filters, Pall Corporation). The second used 1 mL of brines, which was transferred directly to a 2-ml PowerBead tube (MoBio Lab, Inc). Samples were then homogenized using a bead beater with 5 repeated, 1-min cycles. Genomic DNA was subsequently extracted from each treatment for both brines using the PowerBiofilm DNA Isolation kit (MoBio Lab, Inc), following the manufacturer's protocol. The concentration and quality of DNA were measured with a Nanodrop spectrophotometer (Nanodrop 8000, Thermo Scientific). DNA concentration was higher in the filtered treatment for natural brines (8.8 ng/ μ l), and in the direct extraction for concentrated brines (5.6 ng/ μ l); therefore, these treatments were chosen for further analysis. DNA samples were sent to the Research and Testing Laboratory (Lubbock, Texas, USA) for Illumina sequencing. Samples were amplified for sequencing in a two-step process. The forward primer was constructed with (5'-3') the Illumina i5 sequencing primer (TCGTCGGCAGCGTCAG ATGTGTATAAGAGACAG) and the primer 515F (GTGCCAGCMGCCGCGGTAA; Caporaso et al., 2011). The reverse primer was constructed with (5'-3') the Illumina i7 sequencing primer (GTCTCGTGGGCTCGGAGAT GTGTATAAGAGACAG) and the 806R primer (GGACTACHVGGGTWTCTAAT; Caporaso et al., 2011). Amplifications were performed in 25-µl reactions with Qiagen HotStar Tag master mix (Qiagen Inc, Valencia, California), 1 µl of each 5-µM primer, and 1 µl of template. Reactions were performed on ABI Veriti thermocyclers (Applied Biosytems, Carlsbad, California) under the following thermal profile: 95 °C for 5 min, then 35 cycles of 94 °C for 30 s, 54 °C for 40 s, 72 °C for 1 min, followed by one cycle of 72 °C for 10 min, and 4 °C hold. Products from the first stage amplification were added to a second PCR based on gualitatively determine concentrations. Primers for the second PCR were designed based on the Illumina Nextera PCR primers as follows: Forward: AATGATACGGCGACCACCGAGATCTACAC [i5index] TCGTCGGCAGCGTC; Reverse: CAAGCAGAAGACGGCATACGAGAT [i7index]GTCTCGTGGGGCTCGG. The second stage amplification was run the same as the first stage except for 10 cycles. Amplification products were visualized with eGels (Life Technologies, Grand Island, New York). Products were then pooled equimolar and each pool was size selected in two rounds using Agencourt AMPure XP (BeckmanCoulter, Indianapolis, Indiana) in a 0.75 ratio for both rounds. Size selected pools were then quantified using the Quibit 2.0 fluorometer (Life Technologies) and loaded on an Illumina MiSeq (Illumina, Inc. San Diego, California) 2 × 300 flow cell at 10 pM. Raw amplicons reads have been deposited in the Sequence Read Archive of NCBI under accession number SRP091434.

2.4. Data Analyses

Illumina reads were processed using Mothur version 1.27.0 (Schloss et al., 2009), following the standard operational procedure (www.mothur.org/wiki/Miseq_SOP) (Schloss et al., 2011). The screen.seqs command was used to cut off sequences and remove any sequence that was not within the range 100–400 pb. The sequences were aligned with the database SILVA compatible with Mothur. Chimeras were eliminated using UCHIME implemented in Mothur. The commands classify.seqs and remove.lineage were used to identify and remove pollutants classified as mitochondrial, chloroplast, Eukarya, and unknown. In total 47,865 reads from both brines studied were grouped into operational taxonomic units (OTUs), at the 3% level of divergence using the cluster.classic command.





Figure 2. Maucha diagrams characterizing the anionic and cationic content of natural brine, with NaCl concentrations of 255 g/L; and concentrated brine, with LiCl concentrations of 495 g/L.

The exponential of Shannon entropy of the samples was used to calculate the alpha diversity of the samples, according to sample completeness instead of equal sample size (Chao & Jost, 2012) using the iNEXT package in R, to compute and plot seamless rarefaction and extrapolation sampling curves for this diversity index (Hsieh et al., 2016). To test for differences in microbial composition and relative abundances, samples were randomly subsampled to the same size according to the sample with the smallest number of reads. A relaxed neighbor joining phylogenetic tree was constructed using CLEARCUT (Evans et al., 2006) implemented in Mothur and visualized through iTOL (http://itol.embl.de/) (Letunic & Bork, 2011).

The Mothur shared and taxonomy files were converted to a Cytoscape network file using a custom R script (Neave et al., 2014). The data set was used without singletons. The network containing OTUs and samples as nodes, and edges were drawn between OTUs and brines in which they were detected. The weight of the edge was proportional to the abundance of the OTU. The networks were visualized using Cytoscape v3.5.0 (Shannon et al., 2003).

3. Results

3.1. Physicochemical Characteristics

The main ions detected in natural brine were sodium and chloride, while lithium and chloride dominated concentrated brines (Figure 2 and Table S1 in the supporting information) and ion concentrations were markedly different in the two brines. Both brines were circumneutral (natural pH = 6.9; concentrated = 7.3). The brines were both hypersaline (natural salinity = 347; concentrated = 556 g/L; Table 1), total dissolved solids (natural TDS = 348: concentrated = 566 g/L), and total organic carbon (natural TOC = 6.4; concentrated = 8.0 mg/L). Conductivity was 4 times higher in the natural than in the concentrated brines, reflecting the process of solar evaporation, which enriches the brines in solutes (salts) and decreases the amount of water present. Surface tension values were similar in both brines (natural = 80; concentrated = 86 mN/m). The turbidity of the natural brine (2 NTU) was lower than that recorded from concentrated brines (82 NTU), while concentrated brines (26 mPa·s) were more viscous than the natural one (2 mPa·s). The differences in composition and concentrations of ions highlight the effects of the industrial process involved in

Table 1
Physicochemical Parameters of Recorded From Concentrated and Natural Brines From the Salar de Atacama, Chile

Samples	Acidity (g/L)	Conductivity (mS/cm)	Density (g/ml)	Dinamic viscosity (mPa·s)	pН	Salinity (g/L)	Surface tension (mN/m)	Turbidity (NTU)	Total dissolved solids (g/L)	Total organic carbon (mg/L)	Water activity (a _w)
Concentrated brine	2 9.7	59	1.32	26.28	7.03	556.1	86.4	82	566.6	8.0	ND
Natural brine	ND	231	1.22	2.23	6.92	347.3	80.2	< 2	348.8	6.4	0.74

Note. Measuring temperature 22-25 °C. ND not detected.





Figure 3. The Shannon exponential index between samples. The gray shadow shows estimated 95% confidence intervals.

concentrating the brines (Figure S1). There were no detectable concentrations of organic solvents, hydrocarbons, or oils in both brines.

3.2. Differences in Microbial Community Structure Between Brines

After postsequencing cleaning, 591 reads (grouped in 50 OTUs) were classified as Bacteria in the natural brine. A greater number of reads (45,731 grouped in 158 OTUs) was obtained from the concentrated brine. Conversely, in Archaea 1,194 reads (grouped in 193 OTUs) were obtained in the natural brine and 349 reads (grouped in 16 OTUs) from the concentrated brine. Bacterial richness was 3 times higher in the concentrated brine compared with the natural one, but bacterial diversity (Shannon exponential index) was similar in both brines (Figure 3). Archaeal diversity was lower in the concentrated brine (Shannon = 6) compared to natural brine (Shannon = 22). This change in brine microbial community composition was corroborated by quantification of 16S rRNA gene of Archaea through Q-PCR analysis (data not shown). The patterns of microbial richness shown in the samples was reflected by the results of phylogenetic comparisons: bacteria were more phylogenetically diverse in the concentrated brine (Figure 4a) and archaea was more phylogenetically diverse in natural brine (Figure 4b). These results show that natural brines are dominated by Archaea, while Bacteria dominate concentrated brines.

3.3. Microbial Community Composition

Natural and concentrated brines only supported a single halophilic archaeal family: the Halobacteriaceae, which differed in relative abundance at the genus level between the two brines (Figure S3a). This likely reflected differences in the concentration of salts of sodium chloride and lithium chloride. The most abundant archaeal genera (relative abundance) found in natural brine were *Halovenus* (26.8%), *Natronomonas* (20.1%), *Haloarcula* (14%), *Halobacterium* (13%), *Halorientalis* (4.6%), *Halomicroarcula* (1.7%), and *Halorubrum* (1.6%). In addition, the natural brine included a high relative abundance (17%) of unclassified reads in the family Halobacteriaceae. Other genera including *Candidatus Halobonum*, *Halobellus*, and *Halococcus* were present at low relative abundances (<1%).

In the concentrated brine, *Halovenus* was the most abundant archaeal genus (41%), followed by *Haloarcula*, *Halomicroarcula* and *Natronomonas*, which showed lower relative abundance relative to the natural brine. Further, unclassified reads within Halobacteriaceae family reached 2.3% of relative abundance in the concentrated brine. *Halobacterium* and *Halococcus* were the only genera displaying higher abundance in concentrated brine relative to natural brine. This was particularly marked in *Halococcus*, which represented 15% of reads in concentrated brines and only 0.5% in natural brines.







We examined bacterial community composition at family level (Figure S3b), due to the large number of bacterial genera detected. Only 34 families were identified in total and 19 bacterial groups were considered as unclassified at order level. Together, both the Rhodothermaceae (Bacteroidetes) and 2B4 (Gammaproteobacteria) families contributed 65.1% of total bacterial composition in the natural brine. The family Rhodothermaceae (56% relative abundance) was solely represented by the genus *Salinibacter*: the family 2B4 (9.1% relative abundance) was represented by unclassified genera. However, other families also made considerable contributions including an undescribed family, represented by the genus *Thiohalorhabdus* made a contribution of 15.3% in the natural brine. A total of 8.9% was represented by unclassified reads associated with the order Oceanospirillales order, and 2.7% was represented by unclassified bacteria. Other families were detected but at a low relative abundance (<1%), including:



Rhodospirillaceae (0.8%), Family I (Cyanobacteria; 0.7%), Desulfobulbaceae (0.7%), Xanthomonadaceae (0.5%), Chromatiaceae (0.5%), Cyclobacteriaceae (0.3%), Draconibacteriaceae (0.3%), Rhodobacteraceae (0.3%), Flavobacteriaceae (0.3%), Alteromonadaceae (0.2%), Carnobacteriaceae (0.2%), Marinilabiaceae (0.2%), Oceanospirillaceae (0.2%), and Planctomycetaceae (0.2%).

A markedly different bacterial community composition was observed in concentrated brine relative to natural brine, for example, the Rhodothermaceae family was not detected, and the families Xanthomonadaceae (relative abundance = 39.8%) and Staphylococcaceae (26.9%) dominated the bacterial community. Within the Xanthomonadaceae, 99.9% of reads were unclassified; 99.9% of reads associated with the Staphylococcaceae were related with the genus *Staphylococcus*. In addition, unclassified Actinobacteria was detected with 17% of relative abundance. Other families including the Actinomycetaceae, Carnobacteriaceae, Family_XI (Cyanobacteria), and Rhodobacteraceae were detected at low relative abundances (between 1.8% and 2.4%). Fourteen other families were detected from concentrated brine with very low relative abundance (<1%). They were classified as Acetobacteraceae, Bacillaceae, Comamonadaceae, Corynebacteriaceae, Family XIII (Cyanobacteria), Geodermatophilaceae, Litoricolaceae, Microbacteriaceae, Moraxellaceae, Paenibacillaceae, Pseudomonadaceae, Rhodocyclaceae, Solirubrobacteraceae, and Veillonellaceae. Of the reads classified at a family level, 61% were unclassified at genus level.

A comparison of the microbial communities (Archaea and Bacteria) retrieved from the two types of brine showed that only seven OTUs were found in common (Figure S4): all were Archaea (*Halobacterium*, *Natronomonas*, *Halomicroarcula*, *Halorientalis*, *Halococcus*, *Haloarcula*, and *Halovenus*).

4. Discussion

We examined two lithium brines (natural and concentrated brines): they showed markedly different compositions in terms of both their physiochemistry and their microbial communities. The main physicochemical differences between the two brines were viscosity and turbidity, where they were higher in the concentrated brine, giving it a yellow hue and an oily character. Elevated values in these parameters have been associated with high concentrations of salts (Jones & Talley, 1933; Onsager & Samaras, 1934). The Na⁺ and Li⁺ (principal ions of brines) are considered to be salting-out agents, as they decrease the solubility of nonpolar solutes in water (Breslow & Guo, 1990; Kunz et al., 2004) and induce the order in the water structure by increasing the number of hydrogen bonds, which increases the viscosity of the solution (Zangi, 2009). This last effect could be observed in the measurements of viscosity in brines (Table 1). Further, surface tension in both brines (80– 86 mN/m) was high compared to values recorded from seawater (72.93 mN/m; Matubayasi et al., 2001).

The reduction in water availability in hypersaline environments represents a fundamental stress for cellular systems (Hallsworth et al., 2003). The lowest a_w value for growth recorded to date is 0.4 (Steinle et al., 2018). In this study, we observed that natural brine displayed values of a_w of 0.74. However, it was not possible to estimate this parameter from concentrated brine, due to the high concentration of salts (>500 g/L), a value 14 times higher than the mean salt concentration in contemporary oceans (35 g/L; Gargaud, 2011). In any given environment, salinity depends on several factors including the rate of evaporation (more evaporation resulting in an increase in salt concentration) and precipitation (more precipitation resulting in a decrease in salt concentration; Mahajan & Tuteja, 2005). The salinity values obtained in this study was compared with the most hypersaline lakes in the world (Table S2; Arabbeigi et al., 2014; Berube et al., 2007; Bratina et al., 1998; Brock, 1979; Craig, 1969; Dickson et al., 2013; Eder et al., 2001; Haynes & Hammer, 1978; Herbst, 1998; Jiang et al., 2006; Larsen, 1980; Quiroz et al., 2015; Rawson & Moore, 1944; Risacher & Fritz, 1991; Stivaletta et al., 2011). This revealed that natural brine from Salar de Atacama can be considered one of the world's most hypersaline natural environments (dominated by NaCl), in addition to Wadi Natrun Lakes (Zugm, Rizunia; Imhoff et al., 1979) and Lake Assal (Brisou et al., 1974). The concentrated lithium brines of the Salar de Atacama represent one of the most saline environments described to date (dominated by LiCl), followed by the Discovery brine, dominated by MgCl₂ (Yakimov et al., 2015). Furthermore, it has been reported that the lithium chloride shows a very high solubility (820 g/L at 20 °C) (Thies et al., 2002), explaining the extremely high salinities attained in these brines.

An extensive literature exists describing the presence of various taxa living in extremely saline conditions (e.g., Brisou et al., 1974; Imhoff et al., 1979; Pedrós-Alió, 2004; Trüper & Galinski, 1986; Ventosa et al., 1998); however, salinity in these studies typically reflects sodium salts. NaCl-dominated salterns have been reported to be dominated by Euryarchaeota (Kambourova et al., 2016; Maturrano et al., 2006) and

Gammaproteobacteria (Maturrano et al., 2006; Yeon et al., 2005), and to show high archaeal diversity (Andrei et al., 2012; Kambourova et al., 2016), similar to our results, where Euryarchaeota dominate the natural brine. Nevertheless, this does not occur in brines with other chemical compositions. For brines rich in CaCO₃, microbial diversity is typically dominated by Deltaproteobacteria and Alphaproteobacteria (Foti et al., 2008). In the case of a deep-sea brine lake in the Mediterranean Sea rich in MgCl₂, the environment was dominated by Bacteria from the Gammaproteobacteria, Deltaproteobacteria, and Epsilonproteobacteria over Archaea (van der Wielen et al., 2005). In addition, a recent research has detected the presence of the first haloalkaliphilic members of the Candidate Phyla Radiation in soda lake sediments (Na₂CO₃), as well as the dominance of Firmicutes, Bacteroidetes, and Gammaproteobacteria over Archaea (Vavourakis et al., 2018).

It is likely that the relative composition of salts is not the only selective influence on microbial diversity in hypersaline environments but that physical state and water availability (e.g., brine or rock) also play important roles. In the case of the halite-rocks found in the Salar de Atacama, their saline composition (almost entirely NaCl) is similar to the natural brine studied here. However, their microbial composition is characteristically salt adapted, for example, *Cyanobacteria* sp., *Salinibacter* sp., *Halococcus* sp., and *Halorhabdus* sp. (Robinson et al., 2015) and is notably different from the high archaeal diversity seen in the natural brine.

Before this study, little was known regarding the microbiota associated with lithium evaporation ponds in Salar de Atacama. One report suggested that extremely saline (40% and 70% salinity) brines from the Salar de Atacama were devoid of life (Pedrós-Alió, 2004), and another report showed the presence of Archaea but not bacteria in this type of ponds (Demergasso et al., 2004). Recently, Haferburg et al. (2017) examined the microbial diversity in naturally lithium-rich brines from Salar de Uyuni, Bolivia (salinity \leq 356 g/L), only reporting the presence of Archaea in its brines. These microbial groups were also found in our study, but we report the presence of both domains Bacteria and Archaea (Figures 4 and S3), in hypersaline lithium brines of Salar de Atacama (salinity = 556 g/L). The different results regarding the presence of bacteria likely reflect methodological differences, given that the results of high-throughput sequencing depend strongly on primer choice (Klinworth et al., 2013).

The high relative abundance of *Salinibacter* sp. (56%) found in the natural brine (dominated by NaCl) have been widely described, since it is a typical bacterium reported in hypersaline environments (Antón et al., 2000; del Mora-Ruiz et al., 2017; Maturrano et al., 2006). The principal families detected in the concentrated brines were Staphylococcaceae (99.9%) and Xanthomonadaceae-unclassified (99.9%). Some Staphylococcaceae have been reported as displaying halotolerance (Daoud et al., 2013). Furthermore, *Staphylococcus* spp. have been isolated from a hypersaline lake and solar salterns that have NaCl as a principal salt (Brisou et al., 1974; Ghozlan et al., 2006). However, to our understanding, the presence of *Staphylococcus* sp. in lithium hypersaline environments, as well as its halophilic capacity, has not been reported so far. Another striking result was the elevated number of unclassified reads from the family Xanthomonadaceae, as these groups are not typical groups in environments with an extremely high salt concentration, conversely the Xanthomonas strains has been isolated of contaminated soil and waste water (Chang & Zylstra, 2010).

The process of brine concentration via evaporation and subsequent harvesting of salts resulted in a change in the saline composition of the natural brine (high sodium/low lithium concentration) to a brine concentrated in lithium (low sodium/high lithium concentration). In addition, this change resulted in a marked shift in microbial diversity in initial/final brines of the process, with Archaea as the most phylogenetically diverse in natural brine, and Bacteria as the most phylogenetically diverse in concentrated lithium brine (Figures 4a and 4b). However, Archaea was the only group shared in both (Figure S4). We postulate that the factor driving tolerance to changes in salt composition and concentration in these microorganisms is their external structure. Halophilic Archaea such as *Halobacterium salinarum* have a surface layer (*S* layer), composed of single protein or glycoprotein species (Albers & Meyer, 2011). These cellular structures include large amounts of negatively charged amino acids including aspartate and glutamate (Bullock, 2000). When halophiles are present in saline environments such as brines, sodium ions neutralize the negative charges of the carboxyl groups, avoiding cell lysis (Trachtenberg et al., 2000). This could explain how natural brine has a higher richness in Archaea; however, this can be affected when the sodium concentrations decrease in the evaporation ponds, resulting in a decrease of Archaea and a subsequent proportional increase in Bacteria.

In evaporation pond brines, lithium can be considered as a stressing agent acting on microbial diversity selecting against lithium-intolerant microorganisms. This could explain the absence of extremely halophilic



bacteria such as Salinibacter sp. and Halomonas sp. (Haferburg et al., 2017) from concentrated brine. Staphylococcus sp., Xanthomonas sp., and Archaea present in the concentrated brine would likely developed new strategies to survive at high lithium concentrations. In the case of Bacteria, a lithium tolerant isolate classified as Rhodococcus sp. (Urbano et al., 2013) showed the overexpression of enzymes, such as degrader-Protease of misfolded or damaged proteins and proteins related a transcription and translation process, as response of osmotic stress (Belfiore et al., 2017). In parallel, intracellular accumulation of compatible solutes could be a key to survival, reflecting osmotic stress of sodium salt not only in halophiles (Oren, 2010) but also in the case of lithium. Halotolerant and halophilic aerobic bacteria utilize an osmoadaptation "salt-out" strategy, or the accumulation of osmotic solutes to avoid salt stress (Oren et al., 2002). Nevertheless, it has been reported that Salinibacter ruber display a "salt-in" osmoadaptation strategy (similar to the order Halobacteriales; Archaea). In growth conditions, when the concentration of NaCl is greater than 200 g/L, these bacterial cells accumulate high K⁺ and Cl⁻ intracellular concentrations (Oren et al., 2002). This could explain the high abundance of Salinibacter in natural brines (where the concentration and ratio of $Na^+/Cl^$ are high and similar, respectively) and the absence of this genus in concentrated brines, where sodium concentrations decrease and chloride concentrations drastically increase. Furthermore, the salt-in strategy can be considered as an inflexible and very specific methods of osmoadaptation (Welsh, 2000), as all intracellular process must be functional at high NaCl concentrations; therefore, the change in chemical composition of brines (thought evaporation process) could also act as a selective force on the microbial community. The presence of proteins spots for glutamine synthetase, glycerol-3-phosphate dehydrogenase (GDPH), and glycine dehydrogenase in Rhodoccoccus sp., grown in lithium, could reveal information about lithium-resistance strategies, through of accumulation and synthesis of glutamine, glycerol and glycine, as compatible solute and chemical-chaperones for the correct folding of proteins (Belfiore et al., 2017; Diamant et al., 2001). Another possible strategy to survive in high lithium concentrations could be extracellular accumulation of lithium. This is supported by a previous study, which showed that Gram-positive bacteria are able to accumulate the highest concentrations of lithium, unlike gram-negative bacteria, fungi or yeasts, due to the presence of teichoic acids in their cell walls (Tsuruta, 2005). This accumulation of lithium by bacteria could be considered as a mechanism of lithium tolerance, as previously described for other metals (Lemire et al., 2013). However, to date, it has not reported a possible survival strategy to lithium in Archaea.

Microbial life found in contemporary extreme ecosystems can provide information about how early life evolved due to similarities with past environmental conditions (Gutiérrez-Preciado et al., 2018). For example, microbial mats located in the Atacama Desert have been considered as an analogue of major Precambrian environments (Gutiérrez-Preciado et al., 2018). Further, considering the dominant ions in this study (Cl⁻, Li⁺, and Na⁺) and the level of deliquescence of salts and water activity present in the saline systems (Davila et al., 2010), we suggest that each extreme saline environments represent a specific pattern of microbial life that can provide evidence regarding the ecology and geology of ancient ecosystems. Although extremely complicated, attempts to obtain ancient microbial life will likely provide important additional information about survival and adaptative mechanisms, for example, related to ionic composition of natural environments as Antarctic saline lakes (Matsubaya et al., 1979).

The salares (closed saline basins) of northern Chile are under continued threat, largely associated with water extraction to support mining, or the direct exploitation of their salts (Li and B). Salt extraction is in the early stages of development and will increase greatly in the near future to support the demands of commercial and domestic industrial development, especially those related with lithium-ion batteries for electric cars.

It is important to highlight that microbial life—representing the main organisms present in these extreme environments—requires protection within a framework of conservation. Our results clearly demonstrate the uniqueness of Salar de Atacama as an unusual and distinctive habitat for microorganisms. The unique features of these environments are important regarding not only the understanding of the microorganisms inhabiting high LiCl concentrations but also extend to other areas, including astrobiology. For instance, the presence of hydrated (hygroscopic) salts such as magnesium sulfate (Crisler et al., 2012), perchlorates of sodium, and calcium in Martian soils (Ojha et al., 2015) might be key to the development of life.

In recent decades, the Atacama Desert has been subject to an increased focus for astrobiological research, largely due to the presence of "Mars-like" soils (Bull et al., 2016; Navarro-González et al., 2003). These soils harbor active halophilic microbial communities (Ojha et al., 2015; Oren, 2013), specifically in areas with high



concentrations of NaCl, such as halite rocks (Robinson et al., 2015). The presence of abundant deposits of magnesium and sodium salts (major cations), as well as of sulfates and chlorides (major anions) on Mars (Clark & van Hart, 1981; Forte et al., 2016); and the abundant presence of lithium in meteorites (Shima & Honda, 1963) and stars including the Sun (Grevesse, 1968), potentially present important chemical similarities with the brines studied in this research.

Our results showing notably higher bacterial diversity in concentrated brines relative to natural brines, combined with the deliquesce effects of chloride in dry environments (like the Atacama Desert) in supporting an active microbial community in Halite (where at a specific relative humidity the crust is transformed to brines; Davila et al., 2010), have implications for potential life on Mars. It is not too difficult to believe that in deep brines (Orosei et al., 2018) and the chloride-rich deposits of Terra Sirenum (Glotch et al., 2010) of Mars could represent suitable habitat for extant or extinct Bacteria (similar than the concentrated lithium brines). If present, Martian-microbial life probably display a salt-out osmoadaptation strategy, allowing flexibility for variation in ionic composition and water availability.

Consequently, we propose that the brines of Salar de Atacama could represent a useful future model for studies examining biological resistance mechanisms under life-limiting conditions, as well as possible biological indicators for the search and probable development of extraterrestrial life.

References

- Albers, S. V., & Meyer, B. H. (2011). The archaeal cell envelope. *Nature Reviews Microbiology*, 9(6), 414–426. https://doi.org/10.1038/ nrmicro2576
- American Public Health Association. (1915), American Water Works Association, Water Pollution Control Federation, & Water Environment Federation. Standard methods for the examination of water and wastewater (Vol. 2). American Public Health Association.

Andrei, A. Ş., Banciu, H. L., & Oren, A. (2012). Living with salt: Metabolic and phylogenetic diversity of archaea inhabiting saline ecosystems. FEMS Microbiology Letters, 330(1), 1–9. https://doi.org/10.1111/j.1574-6968.2012.02526.x

- Antón, J., Rosselló-Mora, R., Rodríguez-Valera, F., & Amann, R. (2000). Extremely halophilic bacteria in crystallizer ponds from solar salterns. Applied and Environmental Microbiology, 66(7), 3052–3057. https://doi.org/10.1128/AEM.66.7.3052-3057.2000
- Arabbeigi, M., Arzani, A., Majii, M. M., Kiani, R., Tabatabaei, B. E. S., & Habibi, F. (2014). Salinity tolerance of *Aegilops cylindrica* genotypes collected from hyper-saline shores of Uremia Salt Lake using physiological traits and SSR markers. *Acta Physiologiae Plantarum*, 36(8), 2243–2251. https://doi.org/10.1007/s11738-014-1602-0
- Aral, H., & Vecchio-Sadus, A. (2008). Toxicity of lithium to humans and the environment—A literature review. Ecotoxicology and Environmental Safety, 70(3), 349–356. https://doi.org/10.1016/j.ecoenv.2008.02.026
- Belfiore, C., Curia, M. V., & Farías, M. E. (2017). Characterization of Rhodococcus sp. A5 wh isolated from a high altitude Andean lake to unravel the survival strategy under lithium stress. *Revista Argentina de Microbiología*. https://doi.org/10.1016/j.ram.2017.07.005
- Berube, D., Diebel, P., Rollin, A., & Stark, D. (2007). Massive mining evaporation ponds constructed in Chilean desert. Geosynthetics, 25(1), 1-6.
- Bratina, B. J., Stevenson, B. S., Green, W. J., & Schmidt, T. M. (1998). Manganese reduction by microbes from oxic regions of the Lake Vanda (Antarctica) water column. *Applied and Environmental Microbiology*, 64(10), 3791–3797.
- Breslow, R., & Guo, T. (1990). Surface tension measurements show that chaotropic salting-in denaturants are not just water-structure breakers. *Proceedings of the National Academy of Sciences of the United States of America*, 87(1), 167–169.
- Brisou, J., Courtois, D., & Denis, F. (1974). Microbiology Study of a Hypersaline Lake in French Somaliland. Journal of Applied Microbiology, 27(5), 819–822.
- Brock, T. D. (1975). Salinity and the ecology of Dunaliella from Great Salt Lake. *Microbiology*, 89(2), 285–292. https://doi.org/10.1099/00221287-89-2-285
- Brock, T. D. (1979). Ecology of saline lakes. In M. Shilo (Ed.), Strategies of microbial life in extreme environments (pp. 29–47). Berlin: Dahlem Konferenzen.
- Bull, A. T., Asenjo, J. A., Goodfellow, M., & Gómez-Silva, B. (2016). The Atacama Desert: Technical resources and the growing importance of novel microbial diversity. Annual Review of Microbiology, 70, 215–234. https://doi.org/10.1146/annurev-micro-102215-095236
- Bullock, C. (2000). The Archaea—A biochemical perspective. *Biochemistry and Molecular Biology Education*, 28(4), 186–191. https://doi.org/ 10.1111/j.1539-3429.2000.tb00142.x
- Caporaso, J. G., Lauber, C. L., Walters, W. A., Berg-Lyons, D., Lozupone, C. A., Turnbaugh, P. J., et al. (2011). Global patterns of 16S rRNA diversity at a depth of millions of sequences per sample. *Proceedings of the National Academy of Sciences of the United States of America*, 108(Supplement_1), 4516–4522. https://doi.org/10.1073/pnas.1000080107
- Chang, H. K., & Zylstra, G. J. (2010). Xanthomonads. In K. N. Timmis (Ed.), Handbook of hydrocarbon and lipid microbiology. Berlin, Heidelberg: Springer.
- Chao, A., & Jost, L. (2012). Coverage-based rarefaction and extrapolation: Standardizing samples by completeness rather than size. *Ecology*, 93, 2533–2547. https://doi.org/10.1890/11-1952.1

Clark, B. C., & van Hart, D. C. (1981). The salts of Mars. Icarus, 45(2), 370–378. https://doi.org/10.1016/0019-1035(81)90041-5

- Cox, L., Dooleyl, D., & Nestlk, A. (1990). Effect of lithium chloride growth of Listeria spp. International Journal of Food Microbiology, 7, 311–325.
 Craig, H. (1969). Geochemistry and origin of the Red Sea brines. In Hot brines and recent heavy metal deposits in the Red Sea (pp. 368–401).
 Berlin Heidelberg: Springer.
- Crisler, J. D., Newville, T. M., Chen, F., Clark, B. C., & Schneegurt, M. A. (2012). Bacterial growth at the high concentrations of magnesium sulfate found in Martian soils. *Astrobiology*, *12*(2), 98–106. https://doi.org/10.1089/ast.2011.0720
- Daoud, L., Kamoun, J., Ali, M. B., Jallouli, R., Bradai, R., Mechichi, T., et al. (2013). Purification and biochemical characterization of a halotolerant Staphylococcus sp. extracellular lipase. *International Journal of Biological Macromolecules*, 57, 232–237. https://doi.org/10.1016/j. ijbiomac.2013.03.018

Acknowledgments

We gratefully acknowledge Esteban Severino, Patricio Muñoz, Drina Veiar, Vinko Zadjelovic, Vladimir Ávalos and Lucy Muruchi for important comments. Special thanks to Rodrigo Arriaza, Nicolás Miranda, and Chris Harrod for English corrections and figure production to figure production. To Patricia Castillo, Elsa Flores, Teófilo Graber, and Verónica Molina for technical assistance. Finally, our gratitude to Johannes F. Imhoff for his valuable comments that helped us to improve the manuscript and to both reviewers for their useful comments. The data used are listed in section 2. Author Contributions: C. F. C. prepared the samples for analysis, C. F. C. and P. A. run the bioinformatics analysis and wrote most of the manuscript, and M.G. and C. D. contributed with the writing. C. D. and M. G. obtained funding for the project. All authors have read and approved this manuscript. Fundings: This work was supported by a Chilean National Commission for Scientific and Technological Research Scholarship for Doctoral Studies (CONICYT-PCHA/ Doctorado Nacional/2014-21140165) to C. F. C. Further support came from the National Commission for Scientific and Technological Research (FONDECYT 1140179; Centre CeBiB FB 0001) to C. D. and FONDAP Solar Energy Research Center (15110019) to M. G. P. A. is supported by doctoral fellowships from the University of Innsbruck (Austria) and the Corporación Cultural Norte Grande (Universidad de Antofagasta, Chile). The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.



DasSarma, S., & DasSarma, P. (2012). Halophiles. In Encyclopedia of life sciences (pp. 175–186). New York: John Wiley. Davila, A. F., Duport, L. G., Melchiorri, R., Jänchen, J., Valea, S., de los Rios, A., et al. (2010). Hygroscopic salts and the potential for life on Mars. Astrobiology, 10(6), 617–628. https://doi.org/10.1089/ast.2009.0421

de Assunção, L. S., da Luz, J. M. R., da Silva, M. D. C. S., Vieira, P. A. F., Bazzolli, D. M. S., Vanetti, M. C. D., & Kasuya, M. C. M. (2012). Enrichment of mushrooms: An interesting strategy for the acquisition of lithium. *Food Chemistry*, 134(2), 1123–1127. https://doi.org/10.1016/j. foodchem.2012.03.044

Del Barco, R., & Foladori, G. (2014). Nanotechnology and Lithium: A window of opportunity for Bolivia? Advances and Challenges. Paper presented at the 6th International Scientific Conference on Economic and Social Development and 3rd Eastern European ESD Conference: Business Continuity Book of Proceedings: 283, Vienna, Austria.

del Mora-Ruiz, M. R., Cifuentes, A., Font-Verdera, F., Pérez-Fernández, C., Farias, M. E., González, B., et al. (2017). Biogeographical patterns of bacterial and archaeal communities from distant hypersaline environments. *Systematic and Applied Microbiology*, *10*(2), 139–150. https://doi.org/10.1016/j.syapm.2017.10.006

Demergasso, C., Casamayor, E. O., Chong, G., Galleguillos, P., Escudero, L., & Pedrós-Alió, C. (2004). Distribution of prokaryotic genetic diversity in athalassohaline lakes of the Atacama Desert, Northern Chile. FEMS Microbiology Ecology, 48(1), 57–69. https://doi.org/10.1016/ j.femsec.2003.12.013

van der Wielen, P. W. J. J., Bolhuis, H., Borin, S., Daffonchio, D., Corselli, C., Giuliano, L., et al. (2005). The enigma of prokaryotic life in deep hypersaline anoxic basins. *Science*, 307(5706), 121–123. https://doi.org/10.1126/science.1103569

Diamant, S., Eliahu, N., Rosenthal, D., & Goloubinoff, P. (2001). Chemical chaperones regulate molecular chaperones in vitro and in cells under combined salt and heat stresses. Journal of Biological Chemistry, 276(43), 39,586–39,591. https://doi.org/10.1074/jbc.M103081200

Dickson, J. L., Head, J. W., Levy, J. S., & Marchant, D. R. (2013). Don Juan Pond, Antarctica: Near-surface CaCl₂-brine feeding Earth's most saline lake and implications for Mars. *Scientific Reports*, 3(1), 1166. https://doi.org/10.1038/srep01166

Eder, W., Jahnke, L. L., Schmidt, M., & Huber, R. (2001). Microbial diversity of the brine-seawater interface of the Kebrit Deep, Red Sea, studied via 16S rRNA gene sequences and cultivation methods. *Journal of Applied and Environmental Microbiology*, 67(7), 3077–3085. https://doi. org/10.1128/AEM.67.7.3077-3085.2001

Evans, J., Sheneman, L., & Foster, J. (2006). Relaxed neighbor joining: A fast distance-based phylogenetic tree construction method. Journal of Molecular Evolution, 62(6), 785–792. https://doi.org/10.1007/s00239-005-0176-2

Forte, E., Dalle Fratte, M., Azzaro, M., & Guglielmin, M. (2016). Pressurized brines in continental Antarctica as a possible analogue of Mars. Scientific Reports, 6, 33158. https://doi.org/10.1038/srep33158

Foti, M. J., Sorokin, D. Y., Zacharova, E. E., Pimenov, N. V., Kuenen, J. G., & Muyzer, G. (2008). Bacterial diversity and activity along a salinity gradient in soda lakes of the Kulunda Steppe (Altai, Russia). *Extremophiles*, 12(1), 133–145. https://doi.org/10.1007/s00792-007-0117-7

Garcés, I. (2000). Minerales Industriales: Boro, Litio, Salitre y sus derivados. Recursos, Procesos asociados, Mercados y Usos. Registro Nº13602, University de Antofagasta. Chile.

Gargaud, M. (2011). Encyclopedia of astrobiology. France: Springer.

Garrett, D. E. (2004). Handbook of lithium and natural calcium chloride. Amsterdam: Academic Press.

Ghozlan, H., Deif, H., Kandil, R. A., & Sabry, S. (2006). Biodiversity of moderately halophilic bacteria in hypersaline habitats in Egypt. Journal of General and Applied Microbiology, 52(2), 63–72. https://doi.org/10.2323/jgam.52.63

Glotch, T. D., Bandfield, J. L., Tornabene, L. L., Jensen, H. B., & Seelos, F. P. (2010). Distribution and formation of chlorides and phyllosilicates in Terra Sirenum, Mars. *Geophysical Research Letters*, 37, L16202. https://doi.org/10.1029/2010GL044557

Grevesse, N. (1968). Solar abundances of lithium, beryllium and boron. Solar Physics, 5(2), 159–180.

Gruber, P. W., Medina, P. A., Keoleian, G. A., Kesler, S. E., Everson, M. P., & Wallington, T. J. (2011). Global lithium availability. *Journal of Industrial Ecology*, *15*(5), 760–775. https://doi.org/10.1111/j.1530-9290.2011.00359.x

Gutiérrez-Preciado, A., Saghaï, A., Moreira, D., Zivanovic, Y., Deschamps, P., & López-García, P. (2018). Functional shifts in microbial mats recapitulate early Earth metabolic transitions. *Nature Ecology and Evolution*, 1, 1700–1708. https://doi.org/10.1038/s41559-018-0683-3 Habashi, F. (1997). *Handbook of extractive metallurgy*. Germany: Wiley-Vch.

Haferburg, G., Gröning, J. A., Schmidt, N., Kummer, N. A., Erquicia, J. C., & Schlömann, M. (2017). Microbial diversity of the hypersaline and lithium-rich Salar de Uyuni, Bolivia. *Microbiological Research*, 199, 19–28. https://doi.org/10.1016/j.micres.2017.02.007

Hallsworth, J. E., Heim, S., & Timmis, K. N. (2003). Chaotropic solutes cause water stress in Pseudomonas putida. *Environmental Microbiology*, 5(12), 1270–1280. https://doi.org/10.1111/j.1462-2920.2003.00478.x

Haynes, R. C., & Hammer, U. T. (1978). The saline lakes of Saskatchewan IV. Primary production by phytoplankton in selected saline ecosystems. International Review of Hydrobiology, 63(3), 337–351. https://doi.org/10.1002/iroh.19780630304

Herbst, D. B. (1998). Potential salinity limitations on nitrogen fixation in sediments from Mono Lake, California. International Journal of Salt Lake Research, 7(3), 261–274. https://doi.org/10.1007/BF02441878

Hsieh, T. C., Ma, K. H., & Chao, A. (2016). Interpolation and extrapolation for species diversity, R package version 2.0.12. Available at http:// CRAN.R-project.org/package=iNEXT

Imhoff, J. F., Hashwa, F., & Trüper, H. G. (1978). Isolation of extremely halophilic phototrophic bacteria from the alkaline Wadi Natrun, Egypt. Archiv für Hydrobiologie, 84, 381–388.

Imhoff, J. F., Sahl, H. G., Soliman, G. S., & Trüper, H. G. (1979). The Wadi Natrun: Chemical composition and microbial mass developments in alkaline brines of eutrophic desert lakes. *Geomicrobiology Journal*, 1(3), 219–234. https://doi.org/10.1080/01490457909377733

Jiang, H., Dong, H., Zhang, G., Yu, B., Chapman, L. R., & Fields, M. W. (2006). Microbial diversity in water and sediment of Lake Chaka, an athalassohaline lake in northwestern China. *Journal of Applied and Environmental Microbiology*, 72(6), 3832–3845. https://doi.org/10.1128/ AEM.02869-05

Jones, G., & Talley, S. K. (1933). The viscosity of aqueous solutions as a function of the concentration. *Journal of the American Chemical Society*, 55(2), 624–642. https://doi.org/10.1021/ja01329a024

Kambourova, M., Tomova, I., Boyadzhieva, I., Radchenkova, N., & Vasileva-Tonkova, E. (2016). Unusually High Archaeal Diversity in a Crystallizer Pond, Pomorie Salterns, Bulgaria, Revealed by Phylogenetic Analysis. Archaea, 2016, 7459679. https://doi.org/10.1155/2016/ 7459679

Kamekura, M., & Onishi, H. (1982). Cell-associated cations of the moderate halophile Micrococcus varians ssp. halophilus grown in media of high concentrations of LiCl, NaCl, KCl, RbCl, or CsCl. Canadian Journal of Microbiology, 28(2), 155–161. https://doi.org/10.1139/ m82-020

Klinworth, A., Pruesse, E., Schweer, T., Peplies, J., Quast, C., Horn, M., & Glöckner, F. (2013). Evaluation of general 16S ribosomal RNA gene PCR primers for classical and next-generation sequencing-based diversity studies. *Nucleic Acids Research*, *41*(1), e1. https://doi.org/10.1093/ nar/gks808

Kunz, W., Henle, J. W. N. B., & Ninham, B. W. (2004). 'Zur Lehre von der Wirkung der Salze' (about the science of the effect of salts): Franz Hofmeister's historical papers. Current Opinion in Colloid and Interface Science, 9(1), 19–37. https://doi.org/10.1016/j.cocis.2004.05.005 Kurita, N., & Funabashi, M. (1984). Growth-inhibitory effect on fungi of alkali cations and monovalent inorganic anions and antagonism

among different alkali cations. Agricultural and Biological Chemistry, 48(4), 887–893. https://doi.org/10.1271/bbb1961.48.887 Lapierre, L., Undeland, P., & Cox, L. J. (1992). Lithium chloride-sodium propionate agar for the enumeration of bifidobacteria in fermented

dairy products. Journal of Dairy Science, 75(5), 1192–1196. https://doi.org/10.3168/jds.S0022-0302(92)77866-7 Larsen. H. (1980). Ecology of hypersaline environments. Developments in Sedimentology. 28, 23–39. https://doi.org/10.1016/S0070-

- Lemire, J. A., Harrison, J. J., & Turner, R. J. (2013). Antimicrobial activity of metals: Mechanisms, molecular targets and applications. *Nature Reviews Microbiology*, *11*(6), 371–384. https://doi.org/10.1038/nrmicro3028
- Letunic, I., & Bork, P. (2011). Interactive Tree Of Life v2: Online annotation and display of phylogenetic trees made easy. Nucleic Acids Research, 39(suppl_2), 475–478. https://doi.org/10.1093/nar/gkr201
- Lowenstein, T. K., & Risacher, F. (2009). Closed basin brine evolution and the influence of Ca–Cl inflow waters: Death Valley and Bristol Dry Lake California, Qaidam Basin, China, and Salar de Atacama, Chile. *Aquatic Geochemistry*, *15*(1-2), 71–94. https://doi.org/10.1007/s10498-008-9046-z
- Mahajan, S., & Tuteja, N. (2005). Cold, salinity and drought stresses: An overview. Archives of Biochemistry and Biophysics, 444(2), 139–158. https://doi.org/10.1016/j.abb.2005.10.018
- Matsubaya, O., Sakai, H., Torii, T., Burton, H., & Kerry, K. (1979). Antarctic saline lakes—Stable isotopic ratios, chemical compositions and evolution. *Geochimica et Cosmochimica Acta*, 43(1), 7–25. https://doi.org/10.1016/0016-7037(79)90042-5
- Matubayasi, N., Tsunetomo, K., Sato, I., Akizuki, R., Morishita, T., Matuzawa, A., & Natsukari, Y. (2001). Thermodynamic quantities of surface formation of aqueous electrolyte solutions: IV. Sodium halides, anion mixtures, and sea water. *Journal of Colloid and Interface Science*, 243(2), 444–456. https://doi.org/10.1006/jcis.2001.7890
- Maturrano, L., Santos, F., Rosselló-Mora, R., & Antón, J. (2006). Microbial diversity in Maras salterns: A hypersaline environment in the Peruvian Andes. Journal of Applied and Environmental Microbiology, 72(6), 3887–3895. https://doi.org/10.1128/AEM.02214-05
- Minor, T. E., & Marth, E. H. (1976). Staphylococci and their significance in foods. Amsterdam, Oxford: Elsevier Scientific Publishing.
- Mora-Ruiz, M. D. R., Cifuentes, A., Font-Verdera, F., Pérez-Fernández, C., Farias, M. E., González, B., et al. (2018). Biogeographical patterns of bacterial and archaeal communities from distant hypersaline environments. *Systematic and applied microbiology*, 41(2), 139–150.
- Mpodozis, C., Arriagada, C., Basso, M., Roperch, P., Cobbold, P., & Reich, M. (2005). Late Mesozoic to Paleogene stratigraphy of the Salar de Atacama Basin, Antofagasta, Northern Chile: Implications for the tectonic evolution of the Central Andes. *Tectonophysics*, 399(1), 125–154. https://doi.org/10.1016/j.tecto.2004.12.019
- Munk, L. A., Hynek, S. A., Bradley, D., & Jochens, H. (2016). Lithium brines: A global perspective. Economic Geology, 18, 339-365.
- Navarro-González, R., Rainey, F. A., Molina, P., Bagaley, D. R., Hollen, B. J., de la Rosa, J., et al. (2003). Mars-like soils in the Atacama Desert, Chile, and the dry limit of microbial life. *Science*, 302(5647), 1018–1021. https://doi.org/10.1126/science.1089143
- Neave, M., Luter, H., Padovan, A., Townsend, S., Schobben, X., & Gibb, K. (2014). Multiple approaches to microbial source tracking in tropical northern Australia. *Microbiology*, 3(6), 860–874. https://doi.org/10.1002/mbo3.209
- Ochsenreiter, T., Pfeifer, F., & Schleper, C. (2002). Diversity of Archaea in hypersaline environments characterized by molecular-phylogenetic and cultivation studies. *Extremophiles*, 6(4), 267–274. https://doi.org/10.1007/s00792-001-0253-4
- Ojha, L., Wilhelm, M. B., Murchie, S. L., McEwen, A. S., Wray, J. J., Hanley, J., et al. (2015). Spectral evidence for hydrated salts in recurring slope lineae on Mars. *Nature Geoscience*, 8(11), 829–832. https://doi.org/10.1038/ngeo2546
- Onsager, L., & Samaras, N. N. (1934). The surface tension of Debye-Hückel electrolytes. Journal of Chemical Physics, 2(8), 528–536. https://doi. org/10.1063/1.1749522
- Oren, A. (1992). Bacterial activities in the Dead Sea, 1980–-1991: Survival at the upper limit of salinity. International Journal of Salt Lake Research, 1(1), 7–20. https://doi.org/10.1007/BF02904949
- Oren, A. (2010). Industrial and environmental applications of halophilic microorganisms. *Environmental Technology*, 31(8-9), 825–834. https://doi.org/10.1080/09593330903370026
- Oren, A. (2013). Life in magnesium-and calcium-rich hypersaline environments: Salt stress by chaotropic ions. In J. Seckbach, A. Oren, & H. Stan-Lotter (Eds.), *Polyextremophiles: Life Under Multiple Forms of Stress* (pp. 217–232). Dordrecht, Netherlands: Springer.
- Oren, A., Bardavid, R. E., & Mana, L. (2014). Perchlorate and halophilic prokaryotes: Implications for possible halophilic life on Mars. *Extremophiles*, 18(1), 75–80. https://doi.org/10.1007/s00792-013-0594-9
- Oren, A., Heldal, M., Norland, S., & Galinski, E. A. (2002). Intracellular ion and organic solute concentrations of the extremely halophilic bacterium Salinibacter ruber. Extremophiles. 6(6), 491–498. https://doi.org/10.1007/s00792-002-0286-3
- Orosei, R., Lauro, S. E., Pettinelli, E., Cicchetti, A., Coradini, M., Cosciotti, B., et al. (2018). Radar evidence of subglacial liquid water on Mars. *Science*, 361(6401), 490–493. https://doi.org/10.1126/science.aar7268
- Pedrós-Alió, C. (2004). Trophic ecology of solar salterns. In Halophilic microorganisms (pp. 33-48). Berlin, Heidelberg: Springer.
- Perkins, J., & Gadd, G. M. (1993). Accumulation and intracellular compartmentation of lithium ions in Saccharomyces cerevisiae. FEMS Microbiology Letters, 107(2-3), 255–260. https://doi.org/10.1111/j.1574-6968.1993.tb06039.x
- Quiroz, M., Triadó-Margarit, X., Casamayor, E. O., & Gajardo, G. (2015). Comparison of Artemia–bacteria associations in brines, laboratory cultures and the gut environment: A study based on Chilean hypersaline environments. *Extremophiles*, 19(1), 135–147. https://doi.org/ 10.1007/s00792-014-0694-1
- Rawson, D. S., & Moore, J. E. (1944). The saline lakes of Saskatchewan. Canadian Journal of Nursing Research, 22(6), 141–201. https://doi.org/ 10.1139/cjr44d-011
- Risacher, F., & Fritz, B. (1991). Quaternary geochemical evolution of the salars of Uyuni and Coipasa, Central Altiplano, Bolivia. *Chemical Geology*, 90(3-4), 211–231. https://doi.org/10.1016/0009-2541(91)90101-V
- Robinson, C. K., Wierzchos, J., Black, C., Crits-Christoph, A., Ma, B., Ravel, J., et al. (2015). Microbial diversity and the presence of algae in halite endolithic communities are correlated to atmospheric moisture in the hyper-arid zone of the Atacama Desert. *Environmental Microbiology*, 17(2), 299–315. https://doi.org/10.1111/1462-2920.12364
- Sapse, A. M., & Schleyer, P. V. R. (Eds) (1995). Lithium chemistry: A theoretical and experimental overview. New York: John Wiley. Schloss, P. D., Gevers, D., & Westcott, S. L. (2011). Reducing the effects of PCR amplification and sequencing artifacts on 16S rRNA-based studies. PLoS One, 6(12), e27310. https://doi.org/10.1371/journal.pone.0027310
- Schloss, P. D., Westcott, S. L., Ryabin, T., Hall, J. R., Hartmann, M., Hollister, E. B., et al. (2009). Introducing mothur: Open-source, platformindependent, community-supported software for describing and comparing microbial communities. *Applied and Environmental Microbiology*, 75(23), 7537–7541. https://doi.org/10.1128/AEM.01541-09

⁴⁵⁷¹⁽⁰⁸⁾⁷⁰²²⁷⁻⁹

- Shahzad, B., Mughal, M. N., Tanveer, M., Gupta, D., & Abbas, G. (2017). Is lithium biologically an important or toxic element to living organisms? An overview. *Environmental Science and Pollution Research*, 24(1), 103–115. https://doi.org/10.1007/s11356-016-7898-0
 Shannon, P., Markiel, A., Ozier, O., Baliga, N. S., Wang, J. T., Ramage, D., et al. (2003). Cytoscape: A software environment for integrated models
- of biomolecular interaction networks. *Genome Research*, 13(11), 2498–2504. https://doi.org/10.1101/gr.1239303
- Shima, M., & Honda, M. (1963). Isotopic abundance of meteoritic lithium. *Journal of Geophysical Research*, 68(9), 2849–2854. https://doi.org/ 10.1029/JZ068i009p02849
- Steinle, L., Knittel, K., Felber, N., Casalino, C., de Lange, G., Tessarolo, C., et al. (2018). Life on the edge: Active microbial communities in the Kryos MgCl 2-brine basin at very low water activity. *ISME Journal*, 12. https://doi.org/10.1038/s41396-018-0107-z(6), 1414–1426.
- Stivaletta, N., Barbieri, R., Cevenini, F., & Lopez-Garcia, P. (2011). Physicochemical conditions and microbial diversity associated with the evaporite deposits in the Laguna de la Piedra (Salar de Atacama, Chile). *Geomicrobiology Journal, 28*(1), 83–95. https://doi.org/10.1080/01490451003653102
- Sugahara, I., Hayashi, K., Kimura, T., & Jinno, C. (1983). Studies on marine bacteria producing lytic enzymes-IX: Effect of inorganic salts on the autolysis of bacterial cells Capable of Producing Lytic Enzyme. *Bulletin of the Faculty of Fisheries-Mie University (Japan)*, *10*, 33–39.
- Takenishi, M., & Takada, H. (1984). Effect of lithium ions on growth of respiratory deficient mutant of Saccharomyces cerevisiae in medium supplied with galactose as a sole carbón source. International Journal of Antimicrobial Agents, 12, 323–328.
- Thies, H., Nickus, U., Arnold, C., & Psenner, R. (2002). A hydrological tracer experiment with LiCl in a high mountain lake. *Hydrological Processes*, *16*(12), 2329–2337. https://doi.org/10.1002/hyp.1004
- Trachtenberg, S., Pinnick, B., & Kessel, M. (2000). The cell surface glycoprotein layer of the extreme halophile Halobacterium salinarum and its relation to Haloferax volcanii: Cryo-electron tomography of freeze-substituted cells and projection studies of negatively stained envelopes. Journal of Biophysics and Structural Biology, 130(1), 10–26. https://doi.org/10.1006/jsbi.2000.4215
- Trüper, H. G., & Galinski, E. A. (1986). Concentrated brines as habitats for microorganisms. Cellular and Molecular Life Sciences, 42(11), 1182–1187. https://doi.org/10.1007/BF01946388
- Tsuruta, T. (2005). Removal and recovery of lithium using various microorganisms. Journal of Bioscience and Bioengineering, 100(5), 562–566. https://doi.org/10.1263/jbb.100.562
- Urbano, S. B., Albarracín, V. H., Ordoñez, O. F., Farías, M. E., & Alvarez, H. M. (2013). Lipid storage in high-altitude Andean Lakes extremophiles and its mobilization under stress conditions in Rhodococcus sp. A5, a UV-resistant actinobacterium. *Extremophiles*, 17(2), 217–227. https:// doi.org/10.1007/s00792-012-0508-2
- Vavourakis, C. D., Andrei, A. S., Mehrshad, M., Ghai, R., Sorokin, D. Y., & Muyzer, G. (2018). A metagenomics roadmap to the uncultured genome diversity in hypersaline soda lake sediments. *Microbiome*, 6(1), 168. https://doi.org/10.1186/s40168-018-0548-7
- Ventosa, A., de la Haba, R. R., Sánchez-Porro, C., & Papke, R. T. (2015). Microbial diversity of hypersaline environments: A metagenomic approach. Current Opinion in Microbiology, 25, 80–87. https://doi.org/10.1016/j.mib.2015.05.002
- Ventosa, A., Nieto, J. J., & Oren, A. (1998). Biology of moderately halophilic aerobic bacteria. *Microbiology and Molecular Biology Reviews*, 62(2), 504–544.
- Vila, T. (2010). Geología de los depósitos salinos andinos, Provincia de Antofagasta, Chile. Andean Geology, 2. https://doi.org/10.5027/ andgeoV2n1-a05
- Vine, J. D. (1976). Lithium resources and requirements by the year 2000 (No. 1005). United States, US Govt. Print. Off.,
- Wackett, L. P., Dodge, A. G., & Ellis, L. B. (2004). Microbial genomics and the periodic table. Journal of Applied and Environmental Microbiology, 70(2), 647–655. https://doi.org/10.1128/AEM.70.2.647-655.2004
- Warburg, R. J., Moir, A., & Smith, D. A. (1985). Influence of alkali metal cations on the germination of spores of wild-type and gerD mutants of Bacillus subtilis. *Microbiology*, 131(2), 221–230. https://doi.org/10.1099/00221287-131-2-221
- Welsh, D. T. (2000). Ecological significance of compatible solute accumulation by microorganisms: From single cells to global climate. FEMS Microbiology Reviews, 24(3), 263–290. https://doi.org/10.1111/j.1574-6976.2000.tb00542.x
- Yakimov, M. M., la Cono, V., Spada, G. L., Bortoluzzi, G., Messina, E., Smedile, F., et al. (2015). Microbial community of the deep-sea brine Lake Kryos seawater–brine interface is active below the chaotropicity limit of life as revealed by recovery of mRNA. *Environmental Microbiology*, 17(2), 364–382. https://doi.org/10.1111/1462-2920.12587
- Yeon, S. H., Jeong, W. J., & Park, J. S. (2005). The diversity of culturable organotrophic bacteria from local solar salterns. *Journal of Microbiology*, 43(1), 1–10.
- Zangi, R. (2009). Can salting-in/salting-out ions be classified as chaotropes/kosmotropes? *Journal of Physical Chemistry B*, 114(1), 643–650. https://doi.org/10.1021/jp909034c